Author's response to reviews

Title: The effect of gender, age, and geographical location on the incidence and prevalence of renal replacement therapy in Wales

Authors:

Hugo C van Woerden (vanwoerdenh1@cf.ac.uk)
Jane Wilkinson (jane.wilkinson@nphs.wales.nhs.uk)
Martin Heaven (martin.heaven@nphs.wales.nhs.uk)
Jason Merrifield (jason.merrifield@bsc.wales.nhs.uk)

Version: 6 Date: 5 January 2007

Author's response to reviews: see over
Dear Dr Phillips

Re: MS: 1776082852100880 - The effect of gender, age, and geographical location on the incidence and prevalence of renal replacement therapy in Wales

Thank you for your email of 4 January 2007. The following changes have been made to the document.

1. The full name of the organisation commissioning the study and indicating that no ethical approval was required has been more clearly stated.
2. The title page has been amended to conform with your requirements.
3. The abstract has been included and contains 343 words, which is within the required word limit. The relevant headings have been used.
4. The authors checklist for manuscript formatting has been checked against the document. The figures have been separated into individual Word files.

Please let me know if you require any further action.

Kind regards,

Yours sincerely

Dr HC van Woerden